**Author’s response to reviews**

**Title:** Venous thromboembolism and hyperhomocysteinemia as first manifestation of pernicious anaemia: report of four originals cases.

**Authors:**

WAFA AMMOURI (wafaammouri@hotmail.com)

ZOUBIDA TAZI MEZALEK (z.tazimezalek@gmail.com)

HICHAM HARMOUCHE (hharmouche@yahoo.fr)

Mouna MAAAMAR (maamarmouna@yahoo.fr)

Mohamed Adnaoui (adnaouim@yahoo.fr)

**Version:** 1  **Date:** 30 Jun 2017

**Author’s response to reviews:**

Dear reviewers

Thank you for all the comments

I corrected all the grammatical mistakes.

1- All patients had an atrophic gastritis

2- Bone Marrow biopsy was done in three patients and confirmed megaloblastosis. This examen was not done in case 4 because of the absence of hematological abnormalities

3- However, it can be assumed that patients, did not have these deficits, given the clinical presentation of patients and favorable outcome under B12 supplémentation alone.

4- CBS deficiency is characterized by lens dislocation, skeletal abnormalities, neurologic disturbances and thromboembolism. MTHFR deficiency leads to various neurological symptoms, ranging from developpment delay to encephalopathy, including motor and gait abnormalities, seizures, psychiatric manifestations and rarely strokes. The treatment of CBS depends on vitamin B6, whereas MTHFR deficiency can be efficiently treated by vitamin B12, folic acid, and betain